Author's response to reviews

Title: Watchfully checking rapport with the Primary Child Health Care Nurses - a theoretical model from the perspective of parents of foreign origin.

Authors:

   Anita Berlin (anita.berlin@sll.se)
   Lena Törnkvist (lena.tornkvist@sll.se)
   Ingrid Hylander (ingrid.hylander@ki.se)

Version: 2 Date: 6 January 2010

Author's response to reviews:

Dear Editor,

On behalf of my co-authors I hereby have changed the title of the submitted manuscript. As follows: “Watchfully checking rapport with the Primary Child Health Care Nurses - a theoretical model from the perspective of parents of foreign origin”.

All authors certify that they agree on this change of title.

Yours sincerely

Anta Berlin